

Genetic Drift, Purifying Selection and Vector Genotype Shape Dengue Virus Intra-host Genetic Diversity in Mosquitoes

Sebastian Lequime, Albin Fontaine, Meriadeg M. Ar Gouilh, Isabelle Moltini-Conclois, Louis Lambrechts

▶ To cite this version:

Sebastian Lequime, Albin Fontaine, Meriadeg M. Ar Gouilh, Isabelle Moltini-Conclois, Louis Lambrechts. Genetic Drift, Purifying Selection and Vector Genotype Shape Dengue Virus Intra-host Genetic Diversity in Mosquitoes. PLoS Genetics, 2016, 12 (6), pp.e1006111. 10.1371/journal.pgen.1006111. hal-01360140

HAL Id: hal-01360140 https://hal.sorbonne-universite.fr/hal-01360140

Submitted on 5 Sep 2016

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Citation: Lequime S, Fontaine A, Ar Gouilh M, Moltini-Conclois I, Lambrechts L (2016) Genetic Drift, Purifying Selection and Vector Genotype Shape Dengue Virus Intra-host Genetic Diversity in Mosquitoes. PLoS Genet 12(6): e1006111. doi:10.1371/journal.pgen.1006111

Editor: Harmit S. Malik, Fred Hutchinson Cancer Research Center, UNITED STATES

Received: January 5, 2016

Accepted: May 17, 2016

Published: June 15, 2016

Copyright: © 2016 Lequime et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Raw sequences generated in this study were deposited in the NCBI Sequence Read Archive under accession numbers SRP075401 and SRP075335.

Funding: This study was supported by the French Government's Investissement d'Avenir program Laboratoire d'Excellence Integrative Biology of Emerging Infectious Diseases grant ANR-10-LABX-62-IBEID, the City of Paris Emergence(s) program in Biomedical Research, and the French Institut de Recherche Biomédicale des Armées. The funders had no role in study design, data collection and **RESEARCH ARTICLE**

Genetic Drift, Purifying Selection and Vector Genotype Shape Dengue Virus Intra-host Genetic Diversity in Mosquitoes

Sebastian Lequime^{1,2,3}*, Albin Fontaine^{1,2,4}, Meriadeg Ar Gouilh^{5,6}, Isabelle Moltini-Conclois^{1,2}, Louis Lambrechts^{1,2}*

Insect-Virus Interactions Group, Department of Genomes and Genetics, Institut Pasteur, Paris, France,
Centre National de la Recherche Scientifique, Unité de Recherche Associée 3012, Paris, France,
Université Pierre et Marie Curie, Cellule Pasteur UPMC, Paris, France, 4 Equipe Résidente de Recherche d'Infectiologie Tropicale, Division Expertise, Institut de Recherche Biomédicale des Armées, Brétigny-sur-Orge, France, 5 Unité Environnement et Risques Infectieux, Cellule d'Intervention Biologique d'Urgence, Department of Infection and Epidemiology, Institut Pasteur, Paris, France, 6 EA4655, Unité Risques Microbiens U2RM, Université de Caen Normandie, Caen, France

* sebastian.lequime@pasteur.fr (SL); louis.lambrechts@pasteur.fr (LL)

Abstract

Due to their error-prone replication, RNA viruses typically exist as a diverse population of closely related genomes, which is considered critical for their fitness and adaptive potential. Intra-host demographic fluctuations that stochastically reduce the effective size of viral populations are a challenge to maintaining genetic diversity during systemic host infection. Arthropod-borne viruses (arboviruses) traverse several anatomical barriers during infection of their arthropod vectors that are believed to impose population bottlenecks. These anatomical barriers have been associated with both maintenance of arboviral genetic diversity and alteration of the variant repertoire. Whether these patterns result from stochastic sampling (genetic drift) rather than natural selection, and/or from the influence of vector genetic heterogeneity has not been elucidated. Here, we used deep sequencing of full-length viral genomes to monitor the intra-host evolution of a wild-type dengue virus isolate during infection of several mosquito genetic backgrounds. We estimated a bottleneck size ranging from 5 to 42 founding viral genomes at initial midgut infection, irrespective of mosquito genotype, resulting in stochastic reshuffling of the variant repertoire. The observed level of genetic diversity increased following initial midgut infection but significantly differed between mosquito genetic backgrounds despite a similar initial bottleneck size. Natural selection was predominantly negative (purifying) during viral population expansion. Taken together, our results indicate that dengue virus intra-host genetic diversity in the mosquito vector is shaped by genetic drift and purifying selection, and point to a novel role for vector genetic factors in the genetic breadth of virus populations during infection. Identifying the evolutionary forces acting on arboviral populations within their arthropod vector provides novel insights into arbovirus evolution.

analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Author Summary

During infection of their arthropod vectors, arthropod-borne viruses (arboviruses) such as dengue viruses traverse several anatomical barriers that are believed to cause dramatic reductions in population size. Such population bottlenecks challenge the maintenance of viral genetic diversity, which is considered critical for fitness and adaptability of arboviruses. Anatomical barriers in the vector were previously associated with both maintenance of arboviral genetic diversity and alteration of the variant repertoire. However, the relative role of random processes and natural selection, and the influence of vector genetic heterogeneity have not been elucidated. In this study, we used highthroughput sequencing to monitor dengue virus genetic diversity during infection of several genetic backgrounds of their mosquito vector. Our results show that initial infection of the vector is randomly founded by only a few tens of individual virus genomes. The overall level of viral genetic diversity generated during infection was predominantly under purifying selection but differed significantly between mosquito genetic backgrounds. Thus, in addition to random evolutionary forces and the purging of deleterious mutations that shape dengue virus genetic diversity during vector infection, our results also point to a novel role for vector genetic factors in the genetic breadth of virus populations.

Introduction

Due to the low fidelity of their RNA-dependent RNA polymerase, rapid replication kinetics and large population size, RNA viruses consist of a heterogeneous intra-host population of related mutants, sometimes referred to as a quasispecies [1]. This mutant swarm as a whole defines the properties of the viral population, and is considered critical for the fitness and adaptive potential of RNA viruses [1]. For example, high fidelity poliovirus mutants are attenuated in mice *in vivo*, demonstrating the functional importance of intra-host genetic diversity for pathogenesis [2].

Arthropod-borne viruses (arboviruses) are maintained by transmission between vertebrate hosts and blood-feeding arthropods such as mosquitoes that serve as vectors. Although arboviruses span a wide range of viral taxa in the *Togaviridae*, *Flaviviridae*, *Bunyaviridae*, *Rhabdoviridae* and *Orthomyxoviridae* families, the vast majority are RNA viruses, with the single known exception of a DNA arbovirus (African swine fever virus). The genetic plasticity of an RNA genome may confer arboviruses the remarkable ability to alternate between two fundamentally different hosts, and to quickly adapt to novel hosts [3]. Like other RNA viruses, high levels of intra-host genetic diversity are critical for arboviral fitness, as demonstrated in both host types for chikungunya virus [4,5] and West Nile virus [6–8].

Arboviruses usually rely on horizontal transmission between vertebrate hosts and arthropod vectors, although vertical transmission from an infected female arthropod to her offspring may occur [9,10]. After being ingested in a blood meal taken from a viremic vertebrate, arboviruses initially establish infection in the midgut epithelial cells of the arthropod vector. Transmission to another vertebrate host occurs after an extrinsic incubation period during which the arthropod develops a systemic infection that results in the release of viral particles in saliva. During infection of the arthropod vector, arboviruses are confronted with several anatomical barriers that are believed to impose severe population bottlenecks on viral populations [11]. Bottlenecks are dramatic reductions in population size, resulting in stochastic sampling of a small number of viral genomes from the mutant swarm. Population bottlenecks can significantly reduce the

fitness of RNA viruses through accumulation of deleterious mutations that cannot be efficiently removed by purifying selection [12]. Initial infection and traversal of midgut cells, followed by virus dissemination and invasion of the salivary glands are expected to result in strong population drops that represents a challenge to maintaining arboviral genetic diversity during systemic vector infection [13].

Despite such population bottlenecks, arboviruses typically maintain high levels of genetic diversity during transmission by their arthropod vectors [11]. For example, analyses of West Nile virus populations in the midgut, hemolymph and saliva of Culex mosquitoes failed to document reductions in genetic diversity [14]. However, the authors of this study did not determine whether a large effective population size was maintained, or if viral genetic diversity was quickly replenished by mutation and demographic expansion following population bottlenecks. In a recent study of dengue virus (DENV), genetic diversity was maintained during human-to-mosquito transmission but the variant repertoire changed substantially between venous blood and different organs of Aedes mosquitoes that became infected by feeding on the person [15]. Over 90% of DENV genetic variants were lost upon transition from venous blood to mosquito abdomen, as well as from abdomen to salivary glands, which led the authors to estimate that about a hundred viral genomes initially established a productive midgut infection [15]. However, this number could have been underestimated because the calculation assumed that the observed change in variant frequency was due to chance alone (i.e., it did not account for the effect of natural selection). Genetic drift and purifying selection, for example, can result in a similar loss of genetic diversity.

The relative strength of natural selection and genetic drift is informed by the effective population size (N_e) , defined as the size of an idealized population that would drift at the same rate as the observed population [16]. N_e indicates whether the evolution of a population is better described as a deterministic (selection) or stochastic (drift) process. When N_e is large, competition between variants occurs with little interference of random processes. When N_e is small, stochastic sampling of variants counteracts selection and hinders adaptation. For example, genetic drift plays a limited role during systemic infection of the plant host by cauliflower mosaic virus, as viral populations maintain an effective size of several hundreds of viral genomes [17]. Understanding the relative role of genetic drift and natural selection is critical to evaluate the risk of arboviral emergence through adaptive processes [3]. For example, limited epidemic potential of an Asian lineage of chikungunya virus was associated with fixation of a deleterious deletion likely due to a founder effect [18].

In the present study, we investigated the intra-host evolution of DENV in the main mosquito vector *Aedes aegypti* by deep sequencing the full genome of viral populations at different time points of infection. Importantly, we accounted for the potential role of mosquito genetic variation on DENV intra-host genetic diversity. DENV intra-host genetic diversity has attracted considerable interest since the confirmation of its quasispecies nature [<u>19</u>]. Until now, however, most of this research has focused on viral genetic diversity in humans [<u>20–23</u>]. A few studies examined DENV intra-host genetic diversity in the mosquito vector [<u>15,24,25</u>], but these studies did not account for vector genetic heterogeneity. There is substantial evidence for genetic variation in *Ae. aegypti* genotypes and DENV genetic variants [<u>33–37</u>].

Our objectives were three-fold: (*i*) measure the bottleneck size during initial midgut infection of *Ae. aegypti* mosquitoes by DENV; (*ii*) monitor DENV intra-host genetic diversity during population expansion and systemic infection; and (*iii*) determine the influence of the vector genotype on bottleneck size and intra-host DENV genetic diversity.

Material and Methods

Ethics statement

The Institut Pasteur animal facility has received accreditation from the French Ministry of Agriculture to perform experiments on live animals in compliance with the French and European regulations on care and protection of laboratory animals. This study was approved by the Institutional Animal Care and Use Committee at Institut Pasteur.

Virus and mosquitoes

This study used a wild-type DENV-1 isolate (KDH0026A) that was originally recovered from the serum of a clinically ill dengue patient attending Kamphaeng Phet Provincial Hospital, Thailand as previously described [36]. Informed consent of the patient was not necessary because the virus was isolated in laboratory cell culture for diagnostic purposes (unrelated to this study) and, therefore, was no longer a human sample. The isolate was passaged three times in *Aedes albopictus* C6/36 cells prior to its use in this study. The full-length consensus genome sequence of the isolate is available from GenBank under accession number HG316481.

Aedes aegypti females used in this study belonged to the 16th generation of four isofemales lines (referred to as A, B, C, and D thereafter) derived from wild *Ae. aegypti* specimens collected in Kamphaeng Phet Province, Thailand. The lines were initiated by single mating pairs of field-caught males and females as previously described [36]. One male and one female from different collection sites (subdistricts) of the Muang district, Kamphaeng Phet Province, were randomly paired. The mothers of lines A and B, and the father of line C were collected in Thep Na Korn. The fathers of lines A, B and D were collected in Mae Na Ree. The mothers of lines C and D were collected in Nhong Ping Kai. They were maintained in the laboratory by mass sibmating and collective oviposition at each subsequent generation. Quantification of genetic variation within and between the four isofemale lines was conducted as part of this study (see below).

To initiate the experiment, eggs were hatched in filtered tap water. Larvae were reared in $24 \times 34 \times 9$ cm plastic trays at a density of about 200 larvae per tray. Adults were maintained in $30 \times 30 \times 30$ cm screened cages under controlled insectary conditions ($28\pm1^{\circ}$ C, $75\pm5^{\circ}$ relative humidity, 12:12 hour light-dark cycle). They were provided with cotton soaked in a 10% (m/v) sucrose solution *ad libitum* and allowed to mate for 6–7 days before the experimental infection.

Restriction-site associated DNA (RAD) sequencing of mosquitoes

Genetic characterization of the *Ae. aegypti* isofemale lines used single nucleotide polymorphism (SNP) markers identified and genotyped by Restriction-site Associated DNA (RAD) sequencing [<u>38</u>]. Ten females from the 16th generation of each isofemale line (i.e., from the same generation that was used in the experimental infection) and 10 females from the 1st generation of an outbred population collected in 2013 in Thep Na Korn, Kamphaeng Phet Province, Thailand (i.e., the region where the isofemale lines originated) were genotyped using RAD sequencing.

Mosquito genomic DNA was purified using the procedure developed by Pat Roman's laboratory at the University of Toronto [39]. DNA concentration was measured with Qubit fluorometer and Quant-iT dsDNA Assay kit (Life Technologies, Paisley, UK). A modified version of the original double-digest Restriction-site Associated DNA (ddRAD) sequencing protocol [40] was used as previously described [41] with minor additional modifications. Briefly, 350 ng of genomic DNA from each mosquito were digested in a 50-µl reaction containing 50 units each of NlaIII and MluCI restriction enzymes (New England Biolabs, Herts, UK), 1× CutSmart Buffer and water for 3 hours at 37°C, without a heat-kill step. Digestion products were cleaned with 1.5× volume of Ampure XP paramagnetic beads (Beckman Coulter, Brea, CA, USA) and ligated to the modified Illumina P1 and P2 adapters with overhangs complementary to NlaIII and MluCI cutting sites, respectively. Each mosquito was uniquely labeled with a combination of P1 and P2 barcodes of variable lengths to increase library diversity at 5' and 3' ends (S1 Table). Ligation reactions were set up in a 45- μ l volume with 2 μ l of 4 μ M P1 and 12 μ M P2 adapters, 1,000 units of T4 ligase and 1× T4 buffer (New England Biolabs) and were incubated at 16°C overnight. Ligations were heat-inactivated at 65°C for 10 minutes and cooled down to room temperature (20-25°C) in a thermocycler at a rate of 1.5°C per 2 minutes. Adapter-ligated DNA fragments from all individuals were pooled and cleaned with 1.5× bead solution. Fragments from 350 to 440 base pairs (bp) were selected using a Pippin-Prep 2% gel cassette (Sage Sciences, Beverly, MA, USA). Finally, 1 µl of the size-selected DNA was used as a template in a 10- μ l PCR reaction with 5 μ l of Phusion High Fidelity 2× Master mix (New England Biolabs) and 1 µl of 50 µM P1 and P2 primers (S1 Table). To reduce bias due to PCR duplicates, 8 PCR reactions were run in parallel, pooled, and cleaned with a 0.8× bead solution to make the final library. At this step, final libraries were quantified by quantitative PCR using the QPCR NGS Library Quantification Kit (Agilent Technologies, Palo Alto, CA, USA).

Libraries containing multiplexed DNA fragments from 50 mosquitoes were sequenced on an Illumina NextSeq platform using a NextSeq 500 High Output 300 cycles v2 kit (Illumina, San Diego, CA, USA) to obtain 150-bp paired-end reads. An optimized final library concentration of 1.1 pM, spiked with 15% PhiX, was loaded onto the flow cell. Raw sequences were deposited in the NCBI Sequence Read Archive under accession number SRP075401.

RAD markers for mosquito genotyping

A previously developed bash script [41] was used to process raw sequencing reads with minor modifications. Briefly, the DDemux program was used for demultiplexing fastq files according to the P1 and P2 barcodes combinations. Sequence quality scores were automatically converted into Sanger format. Sequences were filtered with FASTX-Toolkit. The first 5 bp (i.e., the restriction enzyme cutting site) and last 11 bp of P1 and P2 reads were trimmed. All reads with Phred scores <25 were discarded. P1 and P2 reads were then matched and unpaired reads were sorted as orphans.

Paired reads were aligned to the reference *Ae. aegypti* genome (AaegL3, February 2016) [42] using Bowtie version 0.12.7 [43]. Parameters for the ungapped alignment included a maximum of three mismatches allowed in the seed, suppression of alignments if more than one reportable alignment existed, and a "try-hard" option to find valid alignments. Orphans were combined with all unaligned paired reads and single-end alignment was attempted. All aligned Bowtie output files were merged per individual and were imported into the Stacks pipeline. A catalog of RAD loci used for SNP discovery was created using the ref_map.pl pipeline in Stacks version 1.37 [44,45]. First, sequences aligned to the same genomic location were stacked together and merged to form loci using Pstacks. Only loci with a sequencing depth \geq 3X per individual were retained. Cstacks was used to create a catalog of consensus loci, merging alleles together and Stacks was used to match all identified loci. The Stacks pipeline identified a total of 899,892 RAD loci. The "populations" module was used to export markers with a sequencing depth \geq 10X that were present in \geq 98% of samples. The mosquito phylogenetic analysis was performed with the resulting subset of 2,321 SNPs, which belonged to 1,319 distinct RAD loci (0.15%).

Phylogenetic analysis of mosquitoes

Phylogenetic trees were constructed using a Bayesian Markov Chain Monte Carlo (MCMC) algorithm, implemented in the BEAST 1.8.3 package [46]. Inferences were produced under the coalescent model (constant size), and under the GTR+G (global time reversible with gamma distribution and no invariable sites) and the HKY+G (Hasegawa-Kishino-Yano) substitution models. Heterozygote positions were considered in calculations by enabling the use of IUPAC code and associated degeneracy within the substitution model. The length of MCMC was set at $3x10^7$ states to obtain Effective Sampling Size (ESS) values >200.

Experimental mosquito infection

Six- to seven-day-old *Ae. aegypti* females were deprived of water and sucrose for 24h prior to the infectious blood meal. The virus stock was diluted in cell culture medium (Leibovitz's L-15 medium + 10% heat-inactivated fetal calf serum + non-essential amino-acids + 0.1% penicillin/ streptomycin + 1% sodium bicarbonate) to reach an expected infectious titer of 3×10^6 focus-forming units (FFU) per mL. One volume of virus suspension was mixed with two volumes of freshly drawn rabbit erythrocytes washed in distilled phosphate-buffered saline (DPBS). After gentle mixing, 2.5 mL of the infectious blood meal was placed in each of several Hemotek membrane feeders (Hemotek Ltd, Blackburn, UK) maintained at 37°C and covered with a piece of desalted porcine intestine as a membrane. Sixty μ L of 0.5 M ATP were added to each feeder as a phagostimulant. Each isofemale line was allowed to feed during two rounds of 15 min on different feeders to ensure randomization of a potential feeder effect. Actual virus titer in the blood meal was measured by standard focus-forming assay in C6/36 cells [33]. After feeding, mosquitoes were cold anesthetized on ice and fully engorged females were transferred to 1-pint cardboard cups. They were incubated under controlled conditions (28±1°C, 75±5% relative humidity, 12:12 hour light-dark cycle) in a climatic chamber.

At 4, 7 and 14 days post exposure (dpe), the midgut of 8-12 individuals from each isofemale line (i.e., biological replicates) were dissected. Midguts were homogenized individually in 140 µL of DPBS + 560 µL of QIAamp Viral RNA Mini Kit lysis buffer (Qiagen, Hilden, Germany) during two rounds of 30 sec at 5,000 rpm in a mixer mill (Precellys 24, Bertin Technologies, Montigny le Bretonneux, France). At 7 and 14 dpe, the legs of midgut-dissected mosquitoes were removed and homogenized as described above. At 14 dpe, the salivary glands of the midgut- and leg-less individuals were harvested and processed as above.

Virus deep sequencing

Total RNA was extracted from mosquito homogenates using QIAamp Viral RNA Mini Kit (Qiagen) and reverse transcribed using Transcriptor High Fidelity cDNA Synthesis Kit (Roche Applied Science, Penzberg, Germany) and a specific reverse primer located at the 3' end of the viral genome (<u>S1 Table</u>). Presence and amount of viral cDNA was assessed by quantitative PCR using the LightCycler DNA Master SyberGreen I kit (Roche Applied Science) and custom primer pairs (<u>S1 Table</u>). Absolute quantification used a standard curve generated with serial dilutions of PCR amplicons of known concentration. Selected samples were amplified by 40 cycles of PCR in 10 overlapping amplicons with Q5 High Fidelity DNA polymerase (New England Biolabs) and custom primer pairs (<u>S1 Table</u>).

PCR products were purified with Agencourt AMPure XP magnetic beads (Beckman Coulter) and their concentration was measured by Quant-iT PicoGreen dsDNA fluorometric quantification (Invitrogen). Equal amounts of each amplicon were pooled by sample and brought to a final concentration of 0.2 ng/ μ L. Multiplexed libraries were prepared using Nextera XT DNA Library Preparation Kit (Illumina) and single-end sequenced on an Illumina NextSeq 500 platform using a high-output 75 cycles v1 kit (Illumina). Sequencing reads were demultiplexed using bcl2fastq v2.15.0 (Illumina). Raw sequences were deposited in the NCBI Sequence Read Archive under accession number SRP075335.

After demultiplexing, reads were trimmed to remove Illumina adaptor sequences using Trimmomatic v0.33 [47] and amplification primers if matching sequences were found on either the 5' or 3' end of the reads using Cutadapt v.1.8.3 [48]. Reads shorter than 32 bp were discarded and remaining reads were then mapped to the reference DENV genome sequence using Bowtie2 v2.1.0 [49]. The alignment file was converted, sorted and indexed using Samtools v0.1.19 [50]. Coverage and sequencing depth were assessed using bedtools v2.17.0 [51]. Single nucleotide variants (SNVs) and their proportion among all reads were called using LoFreq* v2.1.1 [52] and their effect at the amino-acid level assessed by SNPdat v.1.0.5 [53].

Viral genetic diversity analyses

Two sets of SNV markers were used for analyses of genetic diversity and natural selection. The 'full' marker set excluded all nucleotide positions in a given sample that had (*i*) a sequencing depth <500X or (*ii*) where potential sequencing or library preparation artifacts [54] were detected. The 'conservative' marker set excluded all nucleotide positions in all samples that had (*i*) a sequencing depth <500X or (*ii*) where potential sequencing or library preparation artifacts [54] were detected in a least one sample. The conservative marker set minimized the potential bias owing to the unique mutational profile of each nucleotide position. However, because some of the overlapping fragments covering the viral genome could not be successfully amplified in a few samples (S1 Fig), the conservative marker set failed to cover large fractions of the viral genome (S2A Fig). The full marker set, conversely, minimized the potential bias owing to distinct evolutionary properties of the different regions of the viral genome.

Genetic complexity of the viral population was estimated using normalized Shannon entropy (S_n) for each nucleotide site [55]:

$$S_n = \frac{-(p\ln(p)) + ((1-p) \times \ln(1-p))}{\ln(4)}$$

where p is the SNV minor allele frequency at the considered position, and ln(4) corresponds to maximum complexity (i.e., four possible nucleotides at each position). For individual SNVs, S_n values range from 0 to 1. For diallelic SNVs, S_n values range from 0 (no diversity) to 0.5 (maximum complexity, when the two alternative nucleotides are present at equal frequency). For each sample, S_n was averaged over all nucleotide sites included in either the full or the conservative set of SNV markers (i.e., total genome length minus number of excluded positions).

Genetic diversity of the viral population was also estimated using nucleotide diversity at each nucleotide site [56]:

$$\pi = \frac{D}{D-1} \times (1 - (p^2 + (p-1)^2))$$

where *D* is the sequencing depth at the considered position and *p* is the SNV minor allele frequency. Like for S_n , π values for a diallelic SNV range from 0 (no polymorphism) to 0.5 (when the two alternative nucleotides are present at equal frequency). For each sample, π was averaged over all nucleotide sites included in either the full or the conservative set of markers. This index of genetic diversity is less sensitive to low-frequency variants than S_n , due to the lack of log-transformation of the frequencies.

Natural selection assessment

The magnitude and direction of natural selection were assessed using the d_N/d_S ratio, which is the ratio between the number of non-synonymous substitutions per non-synonymous site (d_N) over the number of synonymous substitutions per synonymous site (d_S) of a coding sequence, assuming synonymous substitutions are selectively neutral:

$$d_{\scriptscriptstyle S} = \frac{-3 \times \ln\left(1 - \frac{4 \times \frac{S_d}{S_s}}{3}\right)}{4} \text{ and } d_{\scriptscriptstyle N} = \frac{-3 \times \ln\left(1 - \frac{4 \times \frac{N_d}{N_s}}{3}\right)}{4}$$

where S_d is the number of synonymous substitutions in the sequence, S_s is the number of synonymous sites, N_d is the number of non-synonymous substitutions in the sequence and N_s is the number of non-synonymous sites [57]. A d_N/d_S ratio >1 means that there is an excess of normalized number of non-synonymous substitutions relative to the normalized number of synonymous substitutions and is interpreted as evidence for positive selection (i.e., driving change). A d_N/d_S ratio <1 means that there is an excess of normalized number of synonymous substitutions relative to the normalized number of synonymous substitutions relative to the normalized number of synonymous substitutions relative to the normalized number of non-synonymous substitutions and is interpreted as evidence for negative selection (i.e., acting against change). A d_N/d_S ratio equal to 1 is interpreted as evidence for the absence of natural selection (i.e., neutral evolution).

The d_N/d_S ratio was computed using the Nei-Gojobori method [57] with suggested modifications for high-throughput sequencing data [58]. Briefly, N_d and S_d were calculated for each sample as the sum of SNV frequencies. Mean N_d and S_d were computed for each isofemale line at each time point and used for d_N and d_S calculation, respectively. Numbers of synonymous and non-synonymous sites from the initial population consensus sequence were estimated using MEGA v.7.0.16 [59] by computing the number of 0-, 2-, 3- and 4-fold degenerate sites following the Nei-Gojobori method [57]. The full marker set had a variable number of synonymous and non-synonymous sites depending of the number of nucleotide sites retained or excluded for each sample. The conservative marker set had 328.67 synonymous and 1,481.33 non-synonymous sites for all samples.

Statistical testing

Statistical analyses were performed in the statistical environment R, version 3.2.0 (http://www. <u>r-project.org/</u>) using the packages car [60], plyr [61], ggplot2 [62], stringr [63], reshape2 [64], gridExtra [65], fitdistrplus [66] and boot [67]. In all analyses, the individual mosquito sample was considered a biological unit of replication.

Infection prevalence and cDNA copy numbers were compared among isofemale lines at each time point by pairwise Pearson χ^2 tests and pairwise Wilcoxon signed-rank tests, respectively, followed by a Holm correction of *p*-values for multiple testing.

The proportion of SNVs per position, mean S_n and mean π estimates were compared between the input and later time points using pairwise Wilcoxon signed-rank tests and a Holm p-value adjustment. The proportion of SNVs per position, S_n , π and d_N/d_S estimates in midgut samples were analyzed between 4 and 7 dpe as a function of the combined effects of time point and mosquito genotype using a linear model with an identity link function and a normal error distribution. Model validity was verified with quantile-quantile (Q-Q) plots of residuals and by computing Cook's distance to assess influence of observations. Statistically significant effects (p<0.05) of time point, mosquito genotype and their interactions were determined using type-II analysis of variance. Statistically insignificant interactions were removed from the model, subsequently repeating model validation. Statistical testing of pairwise differences between isofemale lines used the linear regression coefficients. Estimated regression coefficients were extracted and their 95% confidence intervals and *p*-values were calculated based on their standard errors compared to a reference level. Isofemale line A was arbitrarily chosen as the reference level.

Bottleneck size estimation

Following a published method [17], bottleneck size at initial midgut infection was estimated by analyzing the change in frequency distribution of neutral markers between blood meal (initial) and midgut (final) samples. Under the assumption of neutrality (i.e., absence of natural selection), the idealized number of founding genomes (*N*) initiating the midgut infection can be computed as:

$$N = \frac{p(1-p)}{Var(p') - Var(p)}$$

where p is the marker allele frequency in the initial population and p' is the marker allele frequency in the final population [17]. This method considers that changes in the genetic variance between sequential samples result exclusively from genetic drift and therefore requires neutral or quasi-neutral markers.

SNVs that were presumably neutral were selected based on the following set of criteria: (*i*) synonymous change at the third codon position, (*ii*) no significant change in mean frequency between sampling time points, (*iii*) SNV detected in \geq 80% of the five viral input replicates (viral stock and blood meal samples), and (*iv*) mean frequency >0.02 in the input population. Confidence intervals of *N* estimates were computed using a bootstrapping procedure as described in [17]. Briefly, for each bootstrap all individuals were sampled with replacement to calculate *N*. This was repeated 1,000 times to generate a distribution of *N* values and derive 95% confidence intervals corresponding to the 2.5 and 97.5 percentiles of the distribution.

Bottleneck simulation

The effect of the initial midgut infection bottleneck on viral diversity indices was simulated in R based on 100 sampling events from an initial viral population containing 100 independent SNVs. SNV minor allele frequency was randomly drawn from an exponential distribution ($\lambda = 100$). Initial viral population size (equivalent to the infectious dose ingested in the blood meal) was drawn from a normal distribution (mean = 2,000; standard deviation = 200). Bottleneck size was drawn from a normal distribution (mean = 28; standard deviation = 5). Mean S_n and mean π were computed for all samples in the presence or the absence of a detection threshold arbitrarily set at an SNV minor allele frequency of 0.01.

Results

Mosquito genetic variation

A genome-wide set of 2,321 SNPs generated by RAD sequencing was used to genetically characterize the four *Ae. aegypti* isofemale lines (A, B, C, and D). These markers had a sequencing depth \geq 10X per sample and were missing in <2% of samples. An outbred *Ae. aegypti* population from the same geographic location where the lines were created was also genotyped to provide a phylogenetic background. Phylogenetic relationships among individuals from the four isofemale lines and the outbred population were determined with a Bayesian method (Fig 1). As expected, the outbred mosquito population was paraphyletic, reflecting its genetic diversity. Mosquitoes from isofemale lines A and B clustered independently with strong statistical support, confirming their distinct genetic identity. Unexpectedly, mosquitoes from isofemale line



Fig 1. Phylogenetic relationships between Aedes aegypti isofemale lines. Bayesian phylogenetic tree representing the genetic diversity across individuals from the four isofemale lines (A, B, C and D) and from a field-derived outbred population (Pop) from the same geographic location where the lines were created. The phylogenetic analysis was based on a GTR+G substitution model of 2,321 SNPs. Putative populations are depicted in different colors. The scale bar indicates the number of substitutions and posterior probabilities are displayed at relevant nodes.

doi:10.1371/journal.pgen.1006111.g001

C grouped with mosquitoes from isofemale line D within the same clade. This could be the result of relatedness between the parents randomly chosen to initiate the lines, as the mothers of lines C and D came from the same collection site and may have been siblings. Two different substitution models for the phylogenetic reconstruction gave similar clustering patterns. Similar results were also obtained when testing a variable number of markers (allowing from 0% to 30% of missing genotypes) with the same method. Because isofemale lines C and D were not unambiguously assigned to different monophyletic groups, they could not be considered as distinct genotypes and were thus combined in all subsequent analyses. They are jointly referred to as line CD hereafter.

Infection time course and sample selection

Mosquitoes from the three different genotypes (A, B, and CD) were exposed to DENV through an artificial blood meal at a final titer of 1.52×10^6 focus-forming units (FFU)/mL. Assuming a blood meal size of approximately 2 µL, the infectious dose ingested by each mosquito was about 3,000 infectious viral particles. The proportion of mosquitoes that acquired a midgut infection ranged from 75 to 100% and did not differ significantly between time points or isofemale lines (Fig 2A). The proportion of mosquitoes with a DENV infection that disseminated to their legs increased from 10–40% at 7 days post exposure (dpe) to 60–100% at 14 dpe, but the rate of virus dissemination to the legs did not differ significantly between isofemale lines (Fig 2A). However, the proportion of mosquitoes with a disseminated infection in the salivary glands was significantly higher for line CD (87.5%) than for line A (37.5%) and line B (41.7%) at 14 dpe (line A vs. line CD, p = 0.037; line B vs. line CD, p = 0.037). Among infected mosquitoes, viral load ranged from 8.9×10^2 to 2.8×10^6 DENV genome copies per sample with no significant difference between isofemale lines at any of the time points, with the exception of lines B and CD that had significantly different viral loads (p = 0.037) in their salivary glands at 14 dpe (Fig 2B).

Deep sequencing of viral genomes was performed for a subset of 78 infected samples at selected time points (Fig 2B) that were processed individually and treated as biological replicates. Some samples were excluded because their low concentration of viral RNA resulted in unsuccessful RT-PCR amplification. A total of 4, 7 and 13 infected midguts at 4 dpe and 7, 11 and 21 infected midguts at 7 dpe were analyzed for lines A, B, and CD, respectively. Ten infected salivary glands at 14 dpe were analyzed in line CD. In addition, DENV genomes were deep sequenced in the initial viral stock and in four replicates of the infectious blood meal. On average, 3,615,466 sequencing reads per sample aligned to the reference DENV genome. Mean DENV genome coverage with a sequencing depth >500X was 10,594 nucleotides per sample, which represents 98.8% of the 10,718 nucleotides of the total genome length. Mean sequencing depth was 24,212X per sample (S1 Fig).

Patterns of viral genetic diversity

The full set of SNV markers retained for population genetic analyses included an average of 5,843 nucleotide sites across the DENV genome, whereas a more conservative set (see <u>Materials</u> and <u>Methods</u>) was restricted to 1,810 nucleotides (S2A Fig). SNVs of the full marker set were randomly distributed across the genome without obvious mutation hot or cold spot (S3 Fig). A new variant reached consensus level (frequency >0.5) in one midgut sample at 4 dpe and one midgut sample at 7 dpe, but the SNV was different in each case. In salivary glands collected at 14 dpe, new variants reached consensus level at 11 positions, none of which was shared among individuals within or between isofemale lines (S3 Fig). In the more restricted conservative set of markers, no variant reached consensus level at any time point (S2B Fig).





Fig 2. Time course of DENV prevalence and viral load. (A) Bar graphs show the percentage of DENV-infected samples stratified by time point, tissue and isofemale line. Relative numbers of positive samples are indicated above the bars. (B) Box plots show the number of DENV genome copies per infected sample stratified by time point, tissue

and isofemale line. Solid dots represent individual samples selected for deep sequencing and open dots represent samples that were not sequenced. dpe = days post exposure. * p<0.05.

doi:10.1371/journal.pgen.1006111.g002

To determine the effect of initial midgut infection on DENV genetic diversity, a first series of analyses compared viral genetic diversity observed in the input samples with genetic diversity observed at any of the later time points. In the full marker set, initial infection of the midgut was associated with an increase in viral genetic diversity relative to the input (Fig 3) both when measured with normalized Shannon entropy S_n (0 dpe vs. 4 dpe, p = 0.0001; 0 dpe vs. 7 dpe, p = 0.002; 0 dpe vs. 14 dpe, p = 0.003) (Fig 3A) and when measured with nucleotide diversity π (0 dpe vs. 4 dpe, p = 0.0001; 0 dpe vs. 7 dpe, p = 0.0004; 0 dpe vs. 14 dpe, p = 0.003) (Fig 3B). Viral diversity was also significantly higher in the salivary glands at 14 dpe than in the midgut at 7 dpe (S_n : p = 0.012; π : p = 0.012). The proportion of variable sites detected also increased following initial midgut infection (S4 Fig) although differences were only statistically significant between 0 dpe and 4 dpe (p = 0.0029) and between 0 dpe and 14 dpe (p = 0.0067). Similarly, in the conservative set of markers, mosquito infection was associated with a relative increase in viral genetic diversity following initial midgut infection, albeit more modestly due to the smaller number of markers, both when measured with normalized Shannon entropy S_n (0 dpe vs. 4 dpe, p = 0.046; 0 dpe vs. 14 dpe, p = 0.008) (S5B Fig) and when measured with nucleotide diversity π (0 dpe vs. 14 dpe, p = 0.008) (S5C Fig). The proportion of variable sites detected, however, did not differ statistically between time points (S5A Fig).

To evaluate the dynamics of DENV genetic diversity during viral population expansion in the midgut, a second series of analyses compared viral genetic diversity between 4 and 7 dpe, accounting for potential differences between mosquito genotypes. In the full set of markers, both the time point (proportion of variable sites: p = 0.03; S_n : p = 0.04; π : p = 0.04) and the isofemale line (proportion of variable sites: p = 0.0035; S_n : p = 0.0002; π : p = 0.0005) significantly influenced viral genetic diversity. Overall, DENV genetic diversity slightly decreased between 4 dpe and 7 dpe. Isofemale line A displayed significantly higher viral genetic diversity than lines B and CD, for all three indices: proportion of variable sites (p = 0.012 and p = 0.0008, respectively), S_n (p = 0.006 and p = 0.0005 respectively) and π (p = 0.017 and p = 0.0001, respectively). Similar results were obtained with the conservative set of markers. Both the time point (proportion of variable sites: p = 0.01; S_n : p = 0.013; π : p = 0.015) and the isofemale line (proportion of variable sites: p = 0.0011; S_n : p = 0.0006; π : p = 0.0029) significantly influenced viral genetic diversity. Overall, DENV genetic diversity slightly decreased between 4 dpe and 7 dpe. Isofemale line A displayed significantly higher viral genetic diversity than lines B and CD, for all three indices: proportion of variable sites (p = 0.017 and p = 0.0002, respectively), S_n (p = 0.0047 and p = 0.0001, respectively) and π (p = 0.012 and p = 0.0007, respectively).

Natural selection

Based on the full set of SNV markers, d_N/d_S ratios were predominantly negative indicating purifying selection (Fig 3C). There was no statistically significant difference in d_N/d_S ratios between time points or mosquito isofemale lines. Computing d_N/d_S ratios was not possible with the conservative set of markers because the smaller number of SNVs resulted in a large proportion of samples with $d_S = 0$. Analysis of d_N/d_S ratios calculated per isofemale line, however, provided results consistent with predominantly purifying selection using the conservative set of markers. Average d_N/d_S ratios were remarkably similar among lines and time points around 0.2218 (S1 Table).



Fig 3. Observed levels of DENV intra-host genetic diversity and natural selection assessment. (A) Averaged Shannon entropy (S_n) per site over all positions per sample. (B) Averaged nucleotide diversity (π) over all positions per sample. (C) d_N/d_S ratios over all coding positions per sample. The

horizontal, dashed red line represents a d_N/d_S ratio of 1, which is interpreted as evidence for neutral evolution (i.e., absence of natural selection). A d_N/d_S ratio >1 is interpreted as evidence for negative (purifying) selection. Letters above the graph indicate statistically significant pairwise differences between time points. For midgut samples, stars above the bars indicate statistically significant pairwise differences between terms line points.

doi:10.1371/journal.pgen.1006111.g003

PLOS GENETICS

SNV position	Mutation	Position in codon	Amino acid	Viral gene	Initial frequency (mean ± sd)	Final frequency 4 dpe (mean ± sd)	Final frequency 7 dpe (mean ± sd)
1556	$A \to C$	3 rd	L	Е	0.021 ± 0.004	0.026 ± 0.028	0.031 ± 0.023
9950	$\textbf{C} \rightarrow \textbf{A}$	3 rd	Т	NS5	0.117 ± 0.022	0.108 ± 0.059	0.095 ± 0.054
10145	$\textbf{C} \rightarrow \textbf{T}$	3 rd	Т	NS5	0.051 ± 0.011	0.021 ± 0.047	0.043 ± 0.097

dpe = days post exposure; sd = standard deviation.

doi:10.1371/journal.pgen.1006111.t001

Bottleneck size estimates

Three SNVs that complied with criteria of quasi-neutral evolution were selected to estimate the idealized number of founding viral genomes (*N*) initiating the midgut infection based on changes in the variance of their frequency between input and midgut samples (Table 1). Based on the three markers, initial midgut infection was founded by 23–34 genomes when estimated at 4 dpe (Fig 4A) and 5–42 genomes when estimated at 7 dpe (Fig 4B). Collectively, 95% confidence intervals ranged from 2 to 161 founding genomes. *N* estimates and their confidence intervals were consistent across time points, especially for marker at position 1556. For this marker, 4 dpe and 7 dpe data were pooled to compute isofemale line-specific *N* estimates. There were no statistically significant differences among lines in the estimated bottleneck size (Fig 4C), ranging from 83 (95% confidence interval: 52–396) founding genomes for line A, to 23 (9–220) for line B and 33 (16–108) for line CD.

Bottleneck simulations

Simulations were performed to model the effect of population bottlenecks on DENV intra-host genetic diversity. The simulation randomly assigned SNV minor allele frequency, initial viral population size and bottleneck size to explore whether a minimum threshold for SNV detection would alter the observed genetic diversity following a population bottleneck compared to the true genetic diversity. When 100 SNVs were present in the input viral population and no minimum detection threshold was set, mean S_n and π estimated in 100 replicate samples decreased following the bottleneck (Fig 5A). However, when only SNVs with a minor allele frequency >1% were detected, mean S_n and π estimates increased after the bottleneck (Fig 5B).

Discussion

We investigated the evolutionary forces acting on DENV populations within their arthropod vector. Specifically, we evaluated the relative effects of natural selection and genetic drift on DENV intra-host evolution in the midgut of *Ae. aegypti*. In addition, we assessed the influence of vector genetic heterogeneity on intra-host viral genetic diversity. Our results show that DENV intra-host genetic diversity in *Ae. aegypti* is shaped by the combined effects of genetic drift, purifying selection and vector genotype. Reshuffling of the variant repertoire during initial infection of the midgut was associated with a bottleneck size ranging from 5 to 42 founding viral genomes, irrespective of the mosquito genotype. DENV genetic diversity increased



Fig 4. Estimates of bottleneck size at initial midgut infection. The estimated number of founding genomes (*N*) that contribute to initial midgut infection is shown for three markers identified by their position on the DENV genome (1556, 9950, 10145). The three markers are SNVs that are assumed to evolve neutrally or quasi-neutrally. Horizontal bars indicate confidence intervals of *N* estimates computed by bootstrapping. (A) *N* estimates based on samples collected at 4 dpe. (B) *N* estimates based on samples collected at 7 dpe. (C) *N* estimates for each isofemale line obtained for marker 1556 using combined 4 dpe and 7 dpe samples.

doi:10.1371/journal.pgen.1006111.g004

significantly following initial infection, but was restricted by strong purifying selection during DENV population expansion in the midgut. Observed levels of DENV genetic diversity in the midgut differed significantly between mosquito isofemale lines despite a similar bottleneck size at initial infection.

Arboviruses typically maintain high levels of genetic diversity during transmission by their arthropod vectors despite anatomical barriers that often result in severe population drops [11]. Such population bottlenecks have been documented for several arboviruses in their vectors using artificial mutant swarms [68], marked viral clones [13], viral replicons [69] or stochastic simulations based on observed changes in variant frequencies [15]. Although the overall level of arboviral genetic diversity is usually maintained during vector infection [14], the viral variant repertoire can be significantly altered [15,68,70]. Presumably, viral genetic diversity is quickly replenished by mutation and demographic expansion following population bottlenecks [11]. However, whether changes in the viral variant repertoire are due to stochastic sampling (i.e., genetic drift), purifying selection (i.e., removal of variants with lower fitness), or vector genetic heterogeneity combined with specific interactions between vector and virus genotypes [33–37] has remained largely unresolved. Our analysis used neutral or quasi-neutral genetic markers to estimate the effective DENV population size during initial infection of the Ae. aegypti midgut. This approach rules out natural selection and only measures the effect of genetic drift due to random sampling. It is worth noting, however, that true neutral mutation may not exist because even synonymous mutations can have a fitness effect, especially in RNA viruses [71]. Deviation



Fig 5. Simulated effects of a population bottleneck and SNV detection threshold on observed levels of genetic diversity. The simulation considered 100 SNVs present in the input population, which were sampled 100 times (infection events) with randomly assigned SNV frequency, initial viral population size and bottleneck size. In the population sampled following the bottleneck, mean π (A, C) and S_n (B, D) are shown when no frequency detection threshold was set (A, B), and when a 1% frequency detection threshold was set (C, D).

doi:10.1371/journal.pgen.1006111.g005

PLOS

GENETICS

from our assumption of neutrality or quasi-neutrality of the chosen markers may have overestimated the bottleneck size. Indeed, both positive selection and negative selection would likely act to decrease the variance of marker frequency and therefore result in a larger estimate of N_e with our method. Therefore, our conclusion that DENV populations undergo a strong population bottleneck during initial midgut infection should be robust to any undetected departure from neutrality. Moreover, we chose markers whose average frequency was similar before and after the bottleneck, supporting the assumption that they were not under directional selection. Our estimation that initial midgut infection is founded by only a few tens of DENV genomes is consistent with previous estimations for DENV based on stochastic simulations using empirical data [15]. We went one step further by demonstrating that genetic drift, rather than natural selection, is the main evolutionary force underlying this population bottleneck.

Although our estimated bottleneck size is larger than for other RNA viruses during host-tohost transmission [72], it is expected to substantially reduce the genetic breadth of the viral quasispecies [73]. This finding has important implications for DENV evolution in general. A small effective population size means that natural selection will be relatively inefficient during human-to-mosquito transmission. It will prevent adaptive evolution especially if beneficial SNVs are present at low frequencies in the mutant swarm transmitted from the human host [74]. On the other hand, the population bottleneck associated with initial midgut infection may not be small enough to prevent the long-term maintenance of defective viral genomes through complementation by co-infection of host cells with functional viruses. Such a phenomenon was previously documented in the case of a stop-codon mutation that became widespread in DENV populations sampled in Myanmar in 2001 [75]. The frequency of the stop-codon mutation was likely high enough to overcome the effect of population bottlenecks during multiple host-to-host transmission events.

During DENV population expansion following initial midgut infection, natural selection was predominantly negative (i.e., acting against change). Accordingly, the consensus sequence remained unchanged in most of the midgut samples. Only in the salivary glands did several SNVs reach consensus level (frequency >0.5), but with no evidence of evolutionary convergence. As was observed for West Nile virus [68], DENV intra-host genetic diversity in midguts slightly decreased between 4 and 7 dpe. Importantly, we found that overall levels of DENV intra-host genetic diversity differed significantly between distinct mosquito genetic backgrounds. Both the initial bottleneck size and the census size of the viral population did not significantly vary among mosquito genotypes, and thus are unlikely to explain this difference. The mechanistic basis of this finding remains to be determined, but we speculate that viral populations may undergo different selective constraints in different mosquito genotypes. Mosquito genotypes could vary in the intensity of purifying selection (i.e., variation in the efficiency of mechanisms that remove deleterious de novo mutations), but this was not supported by our data. Likewise, the overall lack of positive selection that we observed indicates that it is unlikely to be the underlying mechanism. Alternatively, mosquito genotypes may differ in the level of balancing selection (i.e., mechanisms that act to promote genetic diversity such as negative frequencydependent selection or spatiotemporal fluctuations in the strength and direction of selection). The antiviral RNA interference (RNAi) pathway of mosquitoes was suggested to play a role in viral genetic 'diversification' [76,77], by promoting escape to complementary base-pairing required for RNAi-mediated cleavage. Variation in host factors could also result in differences in viral intra-host genetic diversity through subtle changes in viral RNA-dependent RNA polymerase fidelity [78]. Mutation rates of RNA viruses are not only determined by virus-encoded factors, by also by host-dependent processes. Replicase fidelity of a plant RNA virus was found to differ according to the host type [79]. Replication fidelity in retroviruses can be affected by intracellular dNTP imbalance [80,81]. Viral mutation rate can also be influenced by the expression of host genes, such as cellular deaminases that promote hypermutation in RNA viruses [82-84].

Interestingly, the isofemale line that displayed the lowest level of DENV genetic diversity in the midgut (i.e., line CD) was also associated with the highest prevalence and highest viral load in salivary glands. Because we did not examine viral populations in saliva samples, whether this translates in differences of virus transmission potential is unknown. It is tempting to speculate that the vector competence phenotype relates to the level of viral genetic diversity. Unfortunately, we could not compare DENV intra-host diversity in salivary glands between mosquito isofemale lines because DENV amplification was unsuccessful in two out of three lines due to low template concentration. A recent study found differences in the intra-host genetic diversity of West Nile virus among different species of *Culex* mosquitoes [85]. Here, we provided evidence that such differences exist at the intra-specific level. The potential relationship between viral intra-host genetic diversity and vector competence variation among mosquito genotypes deserves further investigation. It will be interesting to determine in future experiments whether the effect of the vector genotype varies according to the mosquito generation, the virus strain, and/or the specific combinations of mosquito lines and virus strains.

Finally, we introduced a non-exclusive, alternative scenario to the 'diversification' hypothesis that may contribute to explain why the level of arboviral genetic diversity increases despite a population bottleneck. Our proposed scenario is based on the counter-intuitive idea that a strong initial population bottleneck may actually result in a higher observed level of genetic diversity if low-frequency SNVs go undetected for methodological reasons. We used a model based on stochastic simulations to illustrate the effect of a minimum detection threshold of low-frequency SNVs on observed genetic diversity. When all SNVs present were detected regardless of their frequency (i.e., no detection threshold), mean viral genetic diversity indices decreased following a simulated population bottleneck. Conversely, mean genetic diversity indices increased when only SNVs present at a frequency >1% were successfully detected. In our empirical data, it was not possible to ascertain whether SNVs newly detected after the initial population bottleneck resulted from *de novo* mutations or were already present prior to the bottleneck at frequencies lower than the detection threshold. However, our model indicated that a change in the SNV frequency spectrum following the population bottleneck combined with a minimum detection threshold is a potential explanation to the observed increased genetic diversity following the initial bottleneck.

Taken together, our results show that DENV intra-host genetic diversity in the mosquito vector is shaped by stochastic events during initial midgut infection due to a sharp reduction in population size, followed by predominantly purifying selection during population expansion and diversification in the midgut. Differential diversification between mosquito isofemale lines indicates a genetic foundation, but the lack of convergent SNVs does not support the existence of mosquito genotype-specific directional selection. We conclude that the evolution of DENV intra-host genetic diversity in mosquitoes is not only driven by genetic drift and purifying selection, but is also modulated by vector genetic factors. Characterizing the evolutionary forces that govern arboviral genetic diversity contributes to understanding their unique biology and adaptive potential.

Supporting Information

S1 Fig. Sequencing coverage and depth by sample. (PDF)

S2 Fig. Distribution of SNV positions and their mean detected frequencies in the conservative marker set. (A) Bars represent the density of markers retained in the conservative marker set for diversity and natural selection analyses along the DENV reference genome indicated on the *x*-axis. (B) Each dot represents the minor allele frequency of a single SNV along the DENV reference genome indicated on the *x*-axis, averaged over all samples from the same time point in which the SNV was detected. Dot size corresponds to the number of samples from the same time point in which the SNV was detected. The horizontal red dashed line represents a frequency of 0.5 above which a new variant becomes the consensus sequence. SNV distributions are stratified by time point. C = capsid protein, E = envelope glycoprotein, M = membrane glycoprotein, NS1 = non-structural glycoprotein 1; NS2A = non-structural protein 2A; NS2B = non-structural protein 2B; NS3 = non-structural protein 3 (protease/helicase); NS4A = non-structural protein 4A; NS4B = non-structural protein 4B; NS5 = non-structural protein 5 (RNA-dependent RNA polymerase). (PDF)

S3 Fig. Distribution of SNV positions and their mean detected frequencies in the full marker set. Each dot represents the minor allele frequency of a single SNV along the DENV reference genome indicated on the *x*-axis, averaged over all samples from the same time point

and isofemale line in which the SNV was detected. Dot size corresponds to the number of samples from the same time point and isofemale line in which the SNV was detected. The horizontal red dashed line represents a frequency of 0.5 above which a new variant becomes the consensus sequence. SNV distributions are stratified by time point and isofemale line. C = capsid protein, E = envelope glycoprotein, M = membrane glycoprotein, NS1 = non-structural glycoprotein 1; NS2A = non-structural protein 2A; NS2B = non-structural protein 2B; NS3 = non-structural protein 3 (protease/helicase); NS4A = non-structural protein 4A; NS4B = non-structural protein 4B; NS5 = non-structural protein 5 (RNA-dependent RNA polymerase).

(PDF)

S4 Fig. Proportion of variable sites detected in the full marker set. Letters indicate statistically significant pairwise differences between time points. (PDF)

S5 Fig. Observed levels of DENV intra-host genetic diversity using the conservative marker set. (A) Proportion of variable sites detected. (B) Averaged Shannon entropy (S_n) per site over all positions per sample. (C) Averaged nucleotide diversity (π) over all positions per sample; (PDF)

S1 Table. d_N/d_S ratios using the conservative marker set. (DOCX)

S1 File. Primers used for mosquito RAD sequencing and virus deep sequencing. (XLSX)

S2 File. R script used for bottleneck simulations. (TXT)

Acknowledgments

The authors thank Marco Vignuzzi, Kenneth Stapleford, Hervé Blanc, Etienne Patin, Maud Fagny, Guillaume Laval, and all members of the Lambrechts lab for insightful discussions. We thank Laura Dickson for critical reading of an earlier version of the manuscript, Catherine Lallemand for assistance with mosquito rearing, and Gordana Rašić and Igor Filipović for help with mosquito genotyping. We are grateful to Rosmari Rodríguez-Roche for providing some of the PCR primer sequences, and to Alongkot Ponlawat and Thanyalak Fansiri for the initial field collection of mosquitoes. We also thank two anonymous reviewers for their constructive comments, which helped improve an earlier version of the manuscript.

Author Contributions

Conceived and designed the experiments: SL LL. Performed the experiments: SL AF IMC. Analyzed the data: SL AF MAG LL. Wrote the paper: SL AF MAG IMC LL.

References

- 1. Lauring AS, Andino R. Quasispecies Theory and the Behavior of RNA Viruses. Manchester M, editor. PLoS Pathog. 2010; 6.
- Vignuzzi M, Stone JK, Arnold JJ, Cameron CE, Andino R. Quasispecies diversity determines pathogenesis through cooperative interactions in a viral population. Nature. 2006; 439: 344–348. PMID: 16327776

- Coffey LL, Forrester N, Tsetsarkin K, Vasilakis N, Weaver SC. Factors shaping the adaptive landscape for arboviruses: implications for the emergence of disease. Future Microbiol. 2013; 8: 155–176. doi: <u>10.</u> <u>2217/fmb.12.139</u> PMID: <u>23374123</u>
- Coffey LL, Beeharry Y, Bordería AV, Blanc H, Vignuzzi M. Arbovirus high fidelity variant loses fitness in mosquitoes and mice. Proc Natl Acad Sci USA. 2011; 108: 16038–16043. doi: <u>10.1073/pnas.</u> <u>1111650108</u> PMID: <u>21896755</u>
- Rozen-Gagnon K, Stapleford KA, Mongelli V, Blanc H, Failloux A-B, Saleh M-C, et al. Alphavirus mutator variants present host-specific defects and attenuation in mammalian and insect models. PLoS Pathog. 2014; 10: e1003877. doi: 10.1371/journal.ppat.1003877 PMID: 24453971
- Ciota AT, Ngo KA, Lovelace AO, Payne AF, Zhou Y, Shi P-Y, et al. Role of the mutant spectrum in adaptation and replication of West Nile virus. J Gen Virol. 2007; 88: 865–874. PMID: <u>17325359</u>
- Ciota AT, Ehrbar DJ, Van Slyke GA, Willsey GG, Kramer LD. Cooperative interactions in the West Nile virus mutant swarm. BMC Evol Biol. 2012; 12: 58. doi: <u>10.1186/1471-2148-12-58</u> PMID: <u>22541042</u>
- Jerzak GVS, Bernard K, Kramer LD, Shi P-Y, Ebel GD. The West Nile virus mutant spectrum is hostdependant and a determinant of mortality in mice. Virology. 2007; 360: 469–476. PMID: <u>17134731</u>
- 9. Lequime S, Lambrechts L. Vertical transmission of arboviruses in mosquitoes: a historical perspective. Infect Genet Evol. 2014; 28: 681–690. doi: 10.1016/j.meegid.2014.07.025 PMID: 25077992
- Lequime S, Paul RE, Lambrechts L. Determinants of arbovirus vertical transmission in mosquitoes. PLoS Pathog. 2016; 12: e1005548. doi: <u>10.1371/journal.ppat.1005548</u> PMID: <u>27171170</u>
- Forrester NL, Coffey LL, Weaver SC. Arboviral bottlenecks and challenges to maintaining diversity and fitness during mosquito transmission. Viruses. 2014; 6: 3991–4004. doi: <u>10.3390/v6103991</u> PMID: <u>25341663</u>
- Duarte E, Clarke D, Moya A, Domingo E, Holland J. Rapid fitness losses in mammalian RNA virus clones due to Muller's ratchet. Proc Natl Acad Sci USA. 1992; 89: 6015–6019. PMID: <u>1321432</u>
- Forrester NL, Guerbois M, Seymour RL, Spratt H, Weaver SC. Vector-borne transmission imposes a severe bottleneck on an RNA virus population. PLoS Pathog. 2012; 8: e1002897. doi: <u>10.1371/journal.</u> <u>ppat.1002897</u> PMID: <u>23028310</u>
- Brackney DE, Pesko KN, Brown IK, Deardorff ER, Kawatachi J, Ebel GD. West Nile Virus Genetic Diversity is Maintained during Transmission by Culex pipiens quinquefasciatus Mosquitoes. PLoS ONE. 2011; 6.
- 15. Sim S, Aw PPK, Wilm A, Teoh G, Hue KDT, Nguyen NM, et al. Tracking Dengue Virus Intra-host Genetic Diversity during Human-to-Mosquito Transmission. PLoS Negl Trop Dis. 2015; 9: e0004052. doi: 10.1371/journal.pntd.0004052 PMID: 26325059
- 16. Wright S. Evolution in Mendelian Populations. Genetics. 1931; 16: 97–159. PMID: 17246615
- Monsion B, Froissart R, Michalakis Y, Blanc S. Large bottleneck size in Cauliflower Mosaic Virus populations during host plant colonization. PLoS Pathog. 2008; 4: e1000174. doi: <u>10.1371/journal.ppat.</u> <u>1000174</u> PMID: <u>18846207</u>
- Chen R, Wang E, Tsetsarkin KA, Weaver SC. Chikungunya virus 3' untranslated region: adaptation to mosquitoes and a population bottleneck as major evolutionary forces. PLoS Pathog. 2013; 9: e1003591. doi: <u>10.1371/journal.ppat.1003591</u> PMID: <u>24009512</u>
- Wang W-K, Lin SR, Lee CM, King CC, Chang SC. Dengue type 3 virus in plasma is a population of closely related genomes: Quasispecies. J Virol. 2002; 76: 4662–4665. PMID: <u>11932434</u>
- Chao D-Y, King C-C, Wang W-K, Chen W-J, Wu H-L, Chang G-JJ. Strategically examining the fullgenome of dengue virus type 3 in clinical isolates reveals its mutation spectra. Virol J. 2005; 2: 72. PMID: 16120221
- Descloux E, Cao-Lormeau V-M, Roche C, De Lamballerie X. Dengue 1 diversity and microevolution, French Polynesia 2001–2006: connection with epidemiology and clinics. PLoS Negl Trop Dis. 2009; 3: e493. doi: <u>10.1371/journal.pntd.0000493</u> PMID: <u>19652703</u>
- 22. Parameswaran P, Charlebois P, Tellez Y, Nunez A, Ryan EM, Malboeuf CM, et al. Genome-wide patterns of intrahuman dengue virus diversity reveal associations with viral phylogenetic clade and interhost diversity. J Virol. 2012; 86: 8546–8558. doi: 10.1128/JVI.00736-12 PMID: 22647702
- Thai KTD, Henn MR, Zody MC, Tricou V, Nguyet NM, Charlebois P, et al. High-resolution analysis of intrahost genetic diversity in dengue virus serotype 1 infection identifies mixed infections. J Virol. 2012; 86: 835–843. doi: 10.1128/JVI.05985-11 PMID: 22090119
- Lin S-R, Hsieh S-C, Yueh Y-Y, Lin T-H, Chao D-Y, Chen W-J, et al. Study of sequence variation of dengue type 3 virus in naturally infected mosquitoes and human hosts: implications for transmission and evolution. J Virol. 2004; 78: 12717–12721. PMID: <u>15507664</u>

- Sessions OM, Wilm A, Kamaraj US, Choy MM, Chow A, Chong Y, et al. Analysis of Dengue Virus Genetic Diversity during Human and Mosquito Infection Reveals Genetic Constraints. PLoS Negl Trop Dis. 2015; 9: e0004044. doi: <u>10.1371/journal.pntd.0004044</u> PMID: <u>26327586</u>
- Anderson JR, Rico-Hesse R. Aedes aegypti vectorial capacity is determined by the infecting genotype of dengue virus. Am J Trop Med Hyg. 2006; 75: 886–892. PMID: <u>17123982</u>
- Armstrong PM, Rico-Hesse R. Efficiency of dengue serotype 2 virus strains to infect and disseminate in Aedes aegpti. Am J Trop Med Hyg. 2003; 68: 539–544. PMID: <u>12812340</u>
- Bennett KE, Olson KE, Munoz M de L, Fernandez-Salas I, Farfan-Ale JA, Higgs S, et al. Variation in vector competence for dengue 2 virus among 24 collections of Aedes aegypti from Mexico and the United States. Am J Trop Med Hyg. 2002; 67: 85–92. PMID: <u>12363070</u>
- Gubler DJ, Nalim S, Tan R, Saipan H, Sulianti Saroso J. Variation in Susceptibility to Oral Infection with Dengue Viruses Among Geographic Strains of Aedes-Aegypti. Am J Trop Med Hyg. 1979; 28: 1045– 1052. PMID: 507282
- Rosen L, Roseboom LE, Gubler DJ, Lien JC, Chaniotis BN. Comparative Susceptibility of Mosquito Species and Strains to Oral and Parenteral Infection with Dengue and Japanese Encephalitis Viruses. Am J Trop Med Hyg. 1985; 34: 603–615. PMID: <u>2860816</u>
- Tardieux I, Poupel O, Lapchin L, Rodhain F. Variation Among Strains of Aedes-Aegypti in Susceptibility to Oral Infection with Dengue Virus Type-2. Am J Trop Med Hyg. 1990; 43: 308–313. PMID: 2221225
- Vazeille-Falcoz M, Mousson L, Rodhain F, Chungue E, Failloux A-B. Variation in oral susceptibility to dengue type 2 virus of populations of Aedes aegypti from the islands of Tahiti and Moorea, French Polynesia. Am J Trop Med Hyg. 1999; 60: 292–299. PMID: <u>10072154</u>
- Lambrechts L, Chevillon C, Albright RG, Thaisomboonsuk B, Richardson JH, Jarman RG, et al. Genetic specificity and potential for local adaptation between dengue viruses and mosquito vectors. BMC Evol Biol. 2009; 9.
- Lambrechts L. Quantitative genetics of Aedes aegypti vector competence for dengue viruses: towards a new paradigm? Trends Parasitol. 2011; 27: 111–114. doi: <u>10.1016/j.pt.2010.12.001</u> PMID: <u>21215699</u>
- Lambrechts L, Quillery E, Noel V, Richardson JH, Jarman RG, Scott TW, et al. Specificity of resistance to dengue virus isolates is associated with genotypes of the mosquito antiviral gene Dicer-2. Proc R Soc B. 2013; 280: –20122437. doi: 10.1098/rspb.2012.2437 PMID: 23193131
- Fansiri T, Fontaine A, Diancourt L, Caro V, Thaisomboonsuk B, Richardson JH, et al. Genetic Mapping of Specific Interactions between Aedes aegypti Mosquitoes and Dengue Viruses. PLoS Genet. 2013; 9: e1003621. doi: <u>10.1371/journal.pgen.1003621</u> PMID: <u>23935524</u>
- Dickson LB, Sanchez-Vargas I, Sylla M, Fleming K, Black WC. Vector competence in West African Aedes aegypti Is Flavivirus species and genotype dependent. PLoS Negl Trop Dis. 2014; 8: e3153. doi: 10.1371/journal.pntd.0003153 PMID: 25275366
- Miller MR, Dunham JP, Amores A, Cresko WA, Johnson EA. Rapid and cost-effective polymorphism identification and genotyping using restriction site associated DNA (RAD) markers. Genome Res. 2007; 17: 240–248. PMID: <u>17189378</u>
- Black WC, DuTeau NM. RAPD-PCR and SSCP analysis for insect population genetic studies. In: Crampton JM, Beard CB, Louis C, editors. The Molecular Biology of Insect Disease Vectors. London: Chapman & Hall; 1997. pp. 361–373.
- Peterson BK, Weber JN, Kay EH, Fisher HS, Hoekstra HE. Double digest RADseq: an inexpensive method for de novo SNP discovery and genotyping in model and non-model species. PLoS ONE. 2012; 7: e37135. doi: <u>10.1371/journal.pone.0037135</u> PMID: <u>22675423</u>
- Rašić G, Filipović I, Weeks AR, Hoffmann AA. Genome-wide SNPs lead to strong signals of geographic structure and relatedness patterns in the major arbovirus vector, Aedes aegypti. BMC Genomics. 2014; 15: 275. doi: 10.1186/1471-2164-15-275 PMID: 24726019
- Nene V, Wortman JR, Lawson D, Haas B, Kodira C, Tu ZJ, et al. Genome sequence of Aedes aegypti, a major arbovirus vector. Science. 2007; 316: 1718–1723. PMID: <u>17510324</u>
- Langmead B, Trapnell C, Pop M, Salzberg SL. Ultrafast and memory-efficient alignment of short DNA sequences to the human genome. Genome Biol. 2009; 10: R25. doi: <u>10.1186/gb-2009-10-3-r25</u> PMID: <u>19261174</u>
- Catchen JM, Amores A, Hohenlohe P, Cresko W, Postlethwait JH. Stacks: building and genotyping Loci de novo from short-read sequences. G3. 2011; 1: 171–182. doi: <u>10.1534/g3.111.000240</u> PMID: <u>22384329</u>
- Catchen J, Hohenlohe PA, Bassham S, Amores A, Cresko WA. Stacks: an analysis tool set for population genomics. Mol Ecol. 2013; 22: 3124–3140. doi: <u>10.1111/mec.12354</u> PMID: <u>23701397</u>
- Drummond AJ, Rambaut A. BEAST: Bayesian evolutionary analysis by sampling trees. BMC Evol Biol. 2007; 7: 214. PMID: <u>17996036</u>

- Bolger AM, Lohse M, Usadel B. Trimmomatic: a flexible trimmer for Illumina sequence data. Bioinformatics. 2014; 30: 2114–2120. doi: <u>10.1093/bioinformatics/btu170</u> PMID: <u>24695404</u>
- **48.** Martin M. Cutadapt removes adapter sequences from high-throughput sequencing reads. EMBnetjournal. 2011; 17: pp. 10–12.
- Langmead B, Salzberg SL. Fast gapped-read alignment with Bowtie 2. Nat Methods. 2012; 9: 357– U54. doi: 10.1038/nmeth.1923 PMID: 22388286
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The Sequence Alignment/Map format and SAMtools. Bioinformatics. 2009; 25: 2078–2079. doi: <u>10.1093/bioinformatics/btp352</u> PMID: <u>19505943</u>
- Quinlan AR, Hall IM. BEDTools: a flexible suite of utilities for comparing genomic features. Bioinformatics. Oxford University Press; 2010; 26: 841–842.
- 52. Wilm A, Aw PPK, Bertrand D, Yeo GHT, Ong SH, Wong CH, et al. LoFreq: a sequence-quality aware, ultra-sensitive variant caller for uncovering cell-population heterogeneity from high-throughput sequencing datasets. Nucleic Acids Res. 2012; 40: 11189–11201. doi: <u>10.1093/nar/gks918</u> PMID: <u>23066108</u>
- Doran AG, Creevey CJ. Snpdat: Easy and rapid annotation of results from de novo snp discovery projects for model and non-model organisms. BMC Bioinformatics. 2013; 14.
- 54. Costello M, Pugh TJ, Fennell TJ, Stewart C, Lichtenstein L, Meldrim JC, et al. Discovery and characterization of artifactual mutations in deep coverage targeted capture sequencing data due to oxidative DNA damage during sample preparation. Nucleic Acids Res. 2013; 41:–e67. doi: <u>10.1093/nar/gks1443</u> PMID: <u>23303777</u>
- 55. Gregori J, Salicru M, Domingo E, Sanchez A, Esteban JI, Rodriguez-Frias F, et al. Inference with viral quasispecies diversity indices: clonal and NGS approaches. Bioinformatics. 2014; 30: 1104–1111.
- Cornman RS, Boncristiani H, Dainat B, Chen Y, vanEngelsdorp D, Weaver D, et al. Population-genomic variation within RNA viruses of the Western honey bee, Apis mellifera, inferred from deep sequencing. BMC Genomics. 2013; 14: 154. doi: <u>10.1186/1471-2164-14-154</u> PMID: <u>23497218</u>
- Nei M, Gojobori T. Simple methods for estimating the numbers of synonymous and nonsynonymous nucleotide substitutions. Mol Biol Evol. 1986; 3: 418–426. PMID: <u>3444411</u>
- Grubaugh ND, Smith DR, Brackney DE, Bosco-Lauth AM, Fauver JR, Campbell CL, et al. Experimental evolution of an RNA virus in wild birds: evidence for host-dependent impacts on population structure and competitive fitness. PLoS Pathog. 2015; 11: e1004874. doi: <u>10.1371/journal.ppat.1004874</u> PMID: <u>25993022</u>
- Kumar S, Stecher G, Tamura K. MEGA7: Molecular Evolutionary Genetics Analysis version 7.0 for bigger datasets. Mol Biol Evol. 2016: msw054.
- 60. Fox J, Weisberg S. An R Companion to Applied Regression. SAGE; 2011.
- 61. Wickham H. The Split-Apply-Combine Strategy for Data Analysis. J Stat Soft. 2011; 40: 1.
- 62. Wickham H. ggplot2. New York, NY: Springer Science & Business Media; 2009. d
- Wickham H. Simple, Consistent Wrappers for Common String Operations [Internet]. 1st ed. 2015. Available: http://CRAN.R-project.org/package=stringr
- 64. Wickham H. Reshaping data with the reshape package. J Stat Soft. 2007.
- 65. Auguie B. Miscellaneous Functions for "Grid" Graphics. 0 ed. 2015. Available: CRAN.R-project.org/ package = gridExtra
- 66. Delignette-Muller M-L, Dutang C, Pouillot R, Denis J-B. Help to Fit of a Parametric Distribution to Non-Censored or Censored Data. 1st ed.
- 67. Canty A, Ripley B. boot: Bootstrap R (S-Plus) Functions. 1st ed. 2015.
- Ciota AT, Ehrbar DJ, Van Slyke GA, Payne AF, Willsey GG, Viscio RE, et al. Quantification of intrahost bottlenecks of West Nile virus in Culex pipiens mosquitoes using an artificial mutant swarm. Infect Genet Evol. 2012; 12: 557–564. doi: 10.1016/j.meegid.2012.01.022 PMID: 22326536
- Gutiérrez S, Thébaud G, Smith DR, Kenney JL, Weaver SC. Demographics of natural oral infection of mosquitos by Venezuelan equine encephalitis virus. J Virol. 2015; 89: 4020–4022. doi: <u>10.1128/JVI.</u> <u>03265-14</u> PMID: <u>25589654</u>
- Stapleford KA, Coffey LL, Lay S, Bordería AV, Duong V, Isakov O, et al. Emergence and transmission of arbovirus evolutionary intermediates with epidemic potential. Cell Host Microbe. 2014; 15: 706–716. doi: <u>10.1016/j.chom.2014.05.008</u> PMID: <u>24922573</u>
- 71. Cuevas JM, Domingo-Calap P, Sanjuán R. The fitness effects of synonymous mutations in DNA and RNA viruses. Mol Biol Evol. 2012; 29: 17–20. doi: 10.1093/molbev/msr179 PMID: 21771719

- 72. Gutiérrez S, Michalakis Y, Blanc S. Virus population bottlenecks during within-host progression and host-to-host transmission. Curr Opin Virol. 2012; 2: 546–555. doi: <u>10.1016/j.coviro.2012.08.001</u> PMID: <u>22921636</u>
- Escarmis C, Lazaro E, Manrubia SC. Population bottlenecks in quasispecies dynamics. Curr Top Microbiol Immunol. 2006; 299: 141–170. PMID: <u>16568898</u>
- Novella IS, Quer J, Domingo E, Holland JJ. Exponential fitness gains of RNA virus populations are limited by bottleneck effects. J Virol. 1999; 73: 1668–1671. PMID: <u>9882378</u>
- Aaskov J, Buzacott K, Thu HM, Lowry K, Holmes EC. Long-term transmission of defective RNA viruses in humans and Aedes mosquitoes. Science. 2006; 311: 236–238. PMID: <u>16410525</u>
- 76. Brackney DE, Beane JE, Ebel GD. RNAi targeting of West Nile virus in mosquito midguts promotes virus diversification. PLoS Pathog. 2009; 5: e1000502. doi: <u>10.1371/journal.ppat.1000502</u> PMID: <u>19578437</u>
- 77. Brackney DE, Schirtzinger EE, Harrison TD, Ebel GD, Hanley KA. Modulation of flavivirus population diversity by RNA interference. J Virol. 2015; 89: 4035–4039. doi: <u>10.1128/JVI.02612-14</u> PMID: <u>25631077</u>
- Combe M, Sanjuán R. Variation in RNA virus mutation rates across host cells. PLoS Pathog. 2014; 10: e1003855. doi: 10.1371/journal.ppat.1003855 PMID: 24465205
- 79. Pita JS, de Miranda JR, Schneider WL, Roossinck MJ. Environment determines fidelity for an RNA virus replicase. J Virol. 2007; 81: 9072–9077. PMID: <u>17553888</u>
- Bebenek K, Roberts JD, Kunkel TA. The effects of dNTP pool imbalances on frameshift fidelity during DNA replication. J Biol Chem. 1992; 267: 3589–3596. PMID: <u>1371272</u>
- Julias JG, Pathak VK. Deoxyribonucleoside triphosphate pool imbalances in vivo are associated with an increased retroviral mutation rate. J Virol. 1998; 72: 7941–7949. PMID: <u>9733832</u>
- Hajjar AM, Linial ML. Modification of retroviral RNA by double-stranded RNA adenosine deaminase. J Virol. 1995; 69: 5878–5882. PMID: <u>7543593</u>
- Cattaneo R, Schmid A, Eschle D, Baczko K, Meulen ter V, Billeter MA. Biased hypermutation and other genetic changes in defective measles viruses in human brain infections. Cell. 1988; 55: 255–265. PMID: 3167982
- O'Hara PJ, Nichol ST, Horodyski FM, Holland JJ. Vesicular stomatitis virus defective interfering particles can contain extensive genomic sequence rearrangements and base substitutions. Cell. 1984; 36: 915–924. PMID: 6323026
- Grubaugh ND, Weger-Lucarelli J, Murrieta RA, Fauver JR, Garcia-Luna SM, Prasad AN, et al. Genetic Drift during Systemic Arbovirus Infection of Mosquito Vectors Leads to Decreased Relative Fitness during Host Switching. Cell Host Microbe. 2016; 19: 481–492. doi: <u>10.1016/j.chom.2016.03.002</u> PMID: <u>27049584</u>